Author's response to reviews

Title: Multi-segmental Foot Motion during Gait in Children

Authors:

Nicola Portinaro (nicola.portinaro@humanitas.it)
Aberto Leardini (leardini@ior.it)
Artemisia Panou (artemisia.panou@humanitas.it)
Valerio Monzani (ilmoppo@hotmail.it)
Paolo Caravaggi (paolo.caravaggi@ior.it)

Version: 2
Date: 28 March 2014

Author's response to reviews:

Following the editor’s requests, the Ethical approval statement has been added to the Methods section.
Additionally, bullet numbers have been removed from the sections’ headers and "subject" has been replaced with "participant" throughout the manuscript.